Author’s response to reviews

Title: Closing the osteoporosis care gap - Increased osteoporosis awareness among geriatrics and rehabilitation teams.

Authors:

Derek A Haaland (haaland@mcmaster.ca)
Dana R Cohen (dana.cohen@sunnybrook.ca)
Courtney C Kennedy (kennedyc@hhsc.ca)
Nadir A Khalidi (khalidn@mcmaster.ca)
Jonathon D Adachi (jd.adachi@sympatico.ca)
Alexandra Papaioannou (papaioannou@hhsc.ca)

Version: 3 Date: 2 June 2009

Author’s response to reviews:

June 1st, 2009

Dear Committee,

We were grateful to receive your further email suggestions (May 11th, 2009). Attached is the revised manuscript based on these suggestions including:
- shortening of the abstract to 350 words.
- replacement of the respective discussion and conclusions paragraphs with the suggested text.

We look forward to hearing back from you again. Please let us know of any further concerns or suggestions.

Sincerely,

Derek Haaland
on behalf of the study coauthors